Author's response to reviews

Title: Dental caries prevalence, knowledge, attitude and preventive practice on oral health among school children of Chepang communities in Nepal.

Authors:

Lonim Prasai Dixit (lprasai@yahoo.com)
Ajay Shakya (drajayshakya@gmail.com)
Manash Shrestha (manashshrestha@yahoo.com)
Ayush Shrestha (hsuyayush@gmail.com)

Version: 5 Date: 24 January 2013

Author's response to reviews:

Dear Dr Folayan and Dr Paiva,

Thank you very much for your input and the opportunity to revise the article. I have made changes to the article as advised. I sincerely hope that my article can give a picture of dental caries prevalence, experience of oral pain and lack of access to preventive measure in Nepalese school children despite of its limitations.

I am very much willing to re revise the article and sincerely hope to be considered for this opportunity.

Kindest regards,

Lonim.